Author's response to reviews

Title: Sleep study, respiratory mechanics, chemosensitive response and quality of life in morbidly obese patients undergoing bariatric surgery: a prospective, randomized, controlled trail.

Authors:

Luis V F Oliveira (oliveira.lvf@pq.cnpq.br)
Isabella C Aguiar (bellaaguiar@yahoo.com.br)
Raquel P Hirata (raquel_hirata@hotmail.com)
Newton S Faria Junior (nsdfj@yahoo.com.br)
Israel S Reis (ireissantos@uol.com.br)
Luciana M M Sampaio (lucianamalosa@uninove.br)
Claudia S Oliveira (csantos@uninove.br)
Paulo T C Carvalho (ptpaulo@terra.com.br)
Fernando S S Leitao Filho (fernandostudart@uol.com.br)
Lilian C Giannasi (odontogiannasi@uol.com.br)
Lia Azevedo Pinto (liaazevedop@hotmail.com)
Carlos Alberto Malheiros (camalheiros@terra.com.br)
Wilson Rodrigues Freitas Jr (wrfjr@uol.com.br)

Version: 4 Date: 11 October 2011

Author's response to reviews: see over
Title: Sleep study, respiratory mechanics, chemosensitive response and quality of life in morbidly obese patients undergoing bariatric surgery: A prospective, randomized, controlled Trail.
Journal: BMC Surgery
MS ID: 7363195258005093

The BMC Series Editorial Production Team

**Major revisions**

- Additional files: Please remove from the manuscript.

The additional files were modified as requested.

**Minor revisions**

- Structure: Change methods and design to methods/design.

The methods and design were modified as requested.

- Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have made the final corrections and revisions for any typographical errors.

Thank you for your attention.

Isabella Aguiar
Co-author